Author's response to reviews

Title: Does interhospital transfer improve outcome of acute myocardial infarction? A propensity score analysis from the Cardiovascular Cooperative Project

Authors:

John M. Westfall (jack.westfall@uchsc.edu)
Catarina I. Kiefe (ckiefe@uab.edu)
Norman W. Weissman (weissman@uab.edu)
Anthony Goudie (agoudie@uab.edu)
Robert M. Centor (rcentor@uab.edu)
O. Dale Williams (odalew@uab.edu)
Jeroan J. Allison (jallison@uab.edu)

Version: 5 Date: 2 September 2008

Author's response to reviews: see over
Dear Dr. Marlee,

Thank you for thoughtful review and provisional acceptance. Attached is our revised manuscript, “Does interhospital transfer improve outcome of acute myocardial infarction? A propensity score analysis from the Cardiovascular Cooperative Project”.

We have made the final revisions per your request and uploaded the corrected documents to the website. See the notes below on changes made in the revision.

Thank you for the opportunity to publish our manuscript. We believe it is a valid addition to the literature on care of patients with acute myocardial infarction

We look forward to hearing from you.

Sincerely

John M. Westfall, MD,MPH
Associate Dean for Rural Health
University of Colorado Denver
Country - Please remember to include the country in the affiliation details. We have added the country affiliation.

Please delete the title from the head of each page. deleted

Keywords - Please remove these. deleted

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].` completed this change in the references.

Author contributions -> Authors' contributions changed

References - The reference list should contain all authors' names, if there are 30 or below. The term `et al.` should not be used. changed

Tables - Tables must be created using a tables tool. Tables must be divided into rows and columns and be orientated in a portrait page layout, without visible vertical lines. We have used the tables tool and believe we have complied with these requirements. Please let me know if there are any other changes you wish to the tables.